Author's response to reviews

Title: Association of Kidney Function with Inflammatory and Procoagulant Markers in a Diverse Cohort: A Cross-Sectional Analysis from the Multi-Ethnic Study of Atherosclerosis (MESA)

Authors:

Christopher Keller (Christopher.Keller@ucsf.edu)
Ronit Katz (RKatz@u.washington.edu)
Mary Cushman (Mary.Cushman@uvm.edu)
Linda F Fried (Linda.Fried@va.gov)
Michael Shlipak (Michael.Shlipak@ucsf.edu)

Version: 3 Date: 13 July 2008

Author's response to reviews: see over
Date: July 12, 2008

To: Melissa Norton, MD
    Editor-in-Chief, BMC Nephrology

From: Christopher R. Keller, MD
    Michael G. Shlipak, MD MPH

Re: Resubmission of Research Article Manuscript: Association of kidney function with inflammatory and procoagulant markers in a diverse cohort: a cross-sectional analysis from the Multi-Ethnic Study of Atherosclerosis (MESA)

Dear Dr. Norton,

Thank you very much for accepting our manuscript. We have taken great care to revise the manuscript to adhere to the guidelines of BMC Nephrology. Attached is the revised copy of the manuscript. Thank you again for your comments.

Sincerely,

Christopher R. Keller, MD
Michael G. Shlipak, MD MPH